Staple here

MA 226 Solutions to Homework

Name

Last five digits of ID number

Solutions to exercises assigned during the week ending

Discussion section (circle one):

T 3:30–4:30 T 4:30–5:30 W 9–10 W 12–1 W 2–3

Homework format:

1. Solutions must be submitted on 8.5 � 11 inch, white, lined paper. No frizzies on the edges. Three-hole-punch paper is recommended.

2. Pages must be assembled according to the order in which the problems were assigned.

3. Neatness is a consideration when we review this work.